Author’s response to reviews

Title: Using four different clinical tools as predictors for pain after total hip arthroplasty. A prospective cohort study

Authors:

Anja Geisler (agei@regionsjaelland.dk)
Josephine Zachodnik (jzc@regionsjaelland.dk)
Jens Laigaard (jelai@regionsjaelland.dk)
Laura Kruuse (kruuse9@msn.com)
Charlotte Sørensen (cvs@regionsjaelland.dk)
Magnus Sandberg (magnus.sandberg@med.lu.se)
Eva Persson (eva_i.persson@med.lu.se)
Ole Mathiesen (omat@regionsjaelland.dk)

Version: 3 Date: 10 Feb 2020

Author’s response to reviews:

1. Trial registration

-- Please note that trial registration date should be the 'First posted date' i.e. February 20, 2018. As your study was registered after enrollment of the first participant, please also include the words 'retrospectively registered' in this statement at the end of your Abstract.

Author answer:
The manuscript has been corrected as suggested by the editor

2. Consent to participate

-- Please also include the consent information seen in your Methods section in the 'Ethics approval and consent to participate' section of the declarations.

Author answer:
The oral and written consent have been added to the Ethics approval and consent to participate section